Author’s response to reviews

Title: Galectin 3 expression in regional lymph nodes and lymph node metastases of oral squamous cell carcinomas

Authors:

Falk Wehrhan (falk.wehrhan@uk-erlangen.de)
Maike Büttner-Herold (maike.buettner@uk-erlangen.de)
Luitpold Distel (luitpold.distel@uk-erlangen.de)
Jutta Ries (jutta.ries@uk-erlangen.de)
Patrick Moebius (moebius.patrick@gmx.de)
Raimund Preidl (raimund.preidl@uk-erlangen.de)
Carol Geppert (carol.geppert@uk-erlangen.de)
Friedrich Neukam (friedrich.neukam.extern@uk-erlangen.de)
Marco Kesting (marco.kesting@uk-erlangen.de)
Manuel Weber (manuel.weber@uk-erlangen.de)

Version: 2 Date: 26 Jun 2018

Author’s response to reviews:

We included all editorial and review comments.

The similarity analysis also includes the title page including the affiliations as well as invariant parts like authors contributes or declarations. To reduce the degree of similarity, we rephrased and shortened some passages of the “methods” section.

Thank you for considering our paper.

Best regards,

Manuel Weber